Author's response to reviews

Title: Treatment Incidence of and Medical Utilization for Hospitalized Subjects with Pathologic Fractures in Taiwan - Survey of the 2008 National Health Insurance Data

Authors:
Yi-Hui Lee (elishalee@hotmail.com.tw)
Yuan-Nian Hsu (hsu2101@mail.tygh.gov.tw)
I-Liang Yu (yu.i.liang@gmail.com)
Dinh-Van Phan (dvan2707@yahoo.com)
Pesus Chou (pschou@ym.edu.tw)
Chien-Lung Chan (clchan@saturn.yzu.edu.tw)
Nan-Ping Yang (yang.nanping@gmail.com)

Version: 8 Date: 7 September 2011

Author's response to reviews:

Manuscript No.: 2658405554700722
Title: Treatment Incidence of and Medical Utilization for Hospitalized Subjects with Pathologic Fractures in Taiwan - Survey of the 2008 National Health Insurance Data

Dear Editorial Office:

We have performed the final revisions (structure changes, initials in authors’ contributions, update of reference 11) and the manuscript is ready for your publication.

Yang Nan-Ping, MD, MSc, PhD.
( the corresponding author )
2011/09/07, in Taiwan